Author’s response to reviews

Title: Protocol for a randomised controlled trial investigating the effectiveness of an online e-health application compared to attention placebo or sertraline in the treatment of generalised anxiety disorder

Authors:

Helen Christensen (Helen.Christensen@anu.edu.au)
Adam Guastella (aguastella@med.usyd.edu.au)
Andrew Mackinnon (Andrew.Mackinnon@unimelb.edu.au)
Kathleen M Griffiths (Kathy.Griffiths@anu.edu.au)
Claire Eagleson (claire.eagleson@sydney.edu.au)
Phillip Batterham (Philip.Batterham@anu.edu.au)
Kanupriya Kalia (Kanupriya.Kalia@anu.edu.au)
Justin Kenardy (j.kenardy@uq.edu.au)
Kylie Bennett (Kylie.Bennett@anu.edu.au)
Ian B Hickie (ianh@med.usyd.edu.au)

Version: 6 Date: 29 April 2010

Author’s response to reviews: see over
Editors-in-Chief
Trials Journal

Dear Editors-in-Chief,

Please find attached the final revised copy of the study protocol 3145883633542454 entitled ‘Protocol for a randomised controlled trial investigating the effectiveness of an online e-health application compared to attention placebo or sertraline in the treatment of generalised anxiety disorder’.

The following changes were made:

- Page 2- Addition of middle initials for ‘KMG’ and IBH’ next to contact email addresses.
- Page 2- Update of email ‘aguastella@med.usyd.edu.au’ to ‘adam.guastella@sydney.edu.au’.
- Page 15- Addition of full name of PATH study- ‘Personality and Total Health’.
- Page 15- Changed ‘to’ to ‘with’ in following sentence ‘Additionally, the demographic data will be analysed to compare those who responded with the general population’.

Thank you for your review and consideration of this study protocol.

Kind Regards,

Prof Helen Christensen
Email: Helen.Christensen@anu.edu.au
Ph: (+612) 6125 8409